Author's response to reviews

Title: Early invasive vulvar squamous cell carcinoma arising in a woman with vulvar pemphigus vulgaris and systemic lupus erythematosus

Authors:

- Roberto Piccoli (rpiccoli@unina.it)
- Vincenzo D Mandato (dariomandato@virgilio.it)
- Giuseppe Bifulco (giuseppebifulco@hotmail.com)
- Pierluigi Giampaolino (giampigi@virgilio.it)
- Chiara Mignogna (chiaramignogna@gmail.com)
- Michele Davide Mignogna (mmignogna@unina.it)
- Luigi Costagliola (lgcostagliola@libero.it)
- Carmine Nappi (cnappi@unina.it)

Version: 6 Date: 28 May 2010

Author's response to reviews:

Dear Editor,

We revised our paper entitled "Early invasive vulvar squamous cell carcinoma arising in a woman with vulvar pemphigus vulgaris and systemic lupus erythematosus" according to Piura’s suggestions.

Looking forward for your kind reply

Yours sincerely

Vincenzo Dario Mandato